Author’s response to reviews

Title: Implementing a best-practice model of gestational diabetes mellitus care in dietetics: a qualitative study

Authors:
Shelley Wilkinson (shelley.wilkinson@mater.org.au)
Maxine O'Brien (maxine.obrien@health.qld.gov.au)
Sally McCray (sally.mccray@mater.org.au)
Desley Harvey (desley.harvey@health.qld.gov.au)

Version: 2 Date: 11 Jan 2019

Author’s response to reviews:

Thank you for accepting our manuscript.

We have made the recommended changes to the title (and running head); lines 3-4 in the first paragraph and the presentation of %s, as noted by the Reviewer.

We have also ensured the Ethics Committees' names have been included as well as the funding body's relation to the running of the study (not involved).